



Article Can Mergers and Acquisitions Promote Technological Innovation in the New Energy Industry? An Empirical Analysis Based on China's Lithium Battery Industry

Jingyi Zhong ^{1,2}, Weide Chun ^{1,*}, Wu Deng ¹ and Hui Gao ^{3,4}

- ¹ School of Management Science, Chengdu University of Technology, Chengdu 610059, China; zhongjingyizeta@sina.com (J.Z.); dengwu0505@163.com (W.D.)
- ² School of International Education, Tianfu College of Southwest University of Finance and Economics, Chengdu 610000, China
- ³ School of Business, Chengdu University of Technology, Chengdu 610059, China; gaohui@cdut.cn
- ⁴ Energy and Environment Policy Research Center, Chengdu University of Technology, Chengdu 610059, China
- * Correspondence: 18202805172@163.com; Tel.: +86-182-0280-5172

Abstract: The advancement of technological capabilities within lithium battery enterprises crucially facilitates the high-quality development of the new energy industry. This study aims to empirically investigate the impact of mergers and acquisitions (M&A) on the technological innovation capacities of these enterprises, with a specific focus on the lithium battery sector in China. Utilizing data from 34 publicly listed companies spanning the period from 2012 to 2022, we employ the multiperiod double-difference method for comprehensive analysis. Researchers have observed that the implementation of merger and acquisition (M&A) strategies by new energy companies leads to an approximately 1.5 percentage point increase in their technological innovation level. However, the improvement in the green technological innovation level is not significant. After a series of robustness tests, the aforementioned conclusion remains valid. Additionally, with the enhancement of firms' knowledge absorption capacity and regional intellectual property protection, M&A activities can further promote technological innovation in new energy companies and contribute to the enhancement of green technological innovation. Heterogeneity analysis has revealed that technological M&A crucially facilitates the improvement of technological innovation levels among listed companies in the lithium battery industry. Implementing M&A strategies not only benefits the enhancement of firms' technological innovation levels but also significantly fosters green technological innovation. Furthermore, further research has indicated that changes in the level of green technological innovation after the implementation of M&A strategies by new energy companies facilitate the reduction of industrial wastewater and sulfur dioxide emissions. The main innovation of this study, which utilizes new energy companies as the research object, is as follows: it reveals the causal relationship and regulatory mechanism between M&A, technological innovation, and green technological innovation in new energy companies. Furthermore, the study analyzes the mechanism that promotes green technological innovation in new energy companies from the intellectual property protection perspective. Moreover, it assesses the heterogeneous impacts of changes in both technological innovation levels and green technological innovation levels on environmental governance after the implementation of M&A activities.

Keywords: new energy industry; green technology innovation; mergers and acquisitions (M&A); lithium battery

1. Introduction

Technological innovation is not only the key to gaining a competitive advantage for new energy companies but also the core factor in enhancing the high-quality development of the new energy industry. It is crucial for achieving the "carbon peaking" goal in a



Citation: Zhong, J.; Chun, W.; Deng, W.; Gao, H. Can Mergers and Acquisitions Promote Technological Innovation in the New Energy Industry? An Empirical Analysis Based on China's Lithium Battery Industry. *Sustainability* **2023**, *15*, 12136. https://doi.org/10.3390/ su151612136

Academic Editor: Rajesh Kumar Jyothi

Received: 25 July 2023 Revised: 5 August 2023 Accepted: 7 August 2023 Published: 8 August 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). planned and step-by-step manner. As a critical industry in the new energy field, the lithium battery industry fundamentally determines the development of new energy vehicles; thus, it crucially affects the new energy sector. The lithium battery industry relies on technological research and development innovation as its core competitiveness, through which it continuously overcomes key technological limits. The aforementioned industry critically influences the economic and social transition towards a green and low-carbon society. In an increasingly competitive external environment, the following research question is paramount: how can Chinese lithium battery companies enhance their technological innovation capabilities? By answering this question, the green and low-carbon transformation of the modern economy and society can be achieved. The state and relevant departments have also attached immense importance to this issue. Since 2021, the State Council's "14th Five-Year Plan and 2035 Long-term Goal Outline" and the National Development and Reform Commission's "14th Five-Year" New Energy Storage Development Implementation Plan have emphasized the need to enhance the innovation capability of new energy storage technologies such as the lithium battery industry and to actively perform research and development in key technologies for new energy storage. Existing theories indicate that enterprises (i.e., the main body of innovation) can realize technological innovation through two paths, namely closed innovation and open innovation. Mergers and acquisitions (M&A) are a crucial method of open innovation [1]. By implementing M&A, enterprises can obtain the technical knowledge of the target company in a short period and produce innovative synergies. The level of innovation provides a theoretical basis.

In recent years, as the external competitive environment has become increasingly fierce, capital in China's lithium battery industry has been quite active, with frequent M&A. Relevant listed companies not only extend vertically to upstream resources but also focus on the M&A associated with new materials and technologies for lithium batteries [2]. For listed lithium battery-industry companies, M&A began in 2016, the M&A amount exceeded 148.4 billion yuan, and the number of M&A attained 87; this amount will attain a new high in 2022 (>212.2 billion yuan), and the number of M&A will attain 272 pens (Figure 1A). Meanwhile, with the accumulation of M&A in the lithium battery industry, the number of patent applications in the lithium battery industry has exhibited a fluctuating upward trend. In 2016, the number of patent applications was approximately 5700, and it will increase to >7500 in 2022. Although large fluctuations exist, the number generally exhibited an upward trend. Thus, it can be observed that vertical mergers and technology mergers are crucial factors that affect the technological innovation of enterprises, and there may be a positive correlation between them. Can listed companies in China's lithium battery industry enhance their technological innovation capabilities through the implementation of M&A strategies? Which type of M&A are the key factors for facilitating the enhancement of technological innovation capabilities of enterprises? How can new energy companies enhance their level of green technological innovation under the M&A context? What impact does the change in the technological innovation of new energy companies after implementing M&A strategies exert on environmental governance? The answers to the preceding questions can provide empirical evidence for the effective implementation of M&A strategies by new energy companies, which is crucial for enhancing the high-quality development level of the new energy industry.

Aiming at the relationship between M&A and corporate technological innovation, existing studies have further analyzed the impact of M&A in different sectors, such as agriculture, manufacturing, and high-tech industries, on technological innovation based on focusing on the mechanism of M&A affecting corporate technological innovation. It is observed that for different industries, there is heterogeneity in the impact of M&A activities in different industries on corporate technological innovation, which is apparently an inhibitory effect or a promotional effect. In the process of developing the new energy industry, M&A inevitably yields significant implications for technological innovation and green technological innovation in the lithium battery industry. However, there has been limited research on the relationship between M&A and technological innovation in the

lithium battery industry, and the direction of M&A's impact on technological innovation in new energy companies remains uncertain, with unclear underlying mechanisms. Additionally, the environmental governance implications of changes in technological innovation and green technological innovation in new energy companies after M&A activities are not comprehensively understood, which highlights the urgent need for relevant studies.



Figure 1. China's lithium battery industry: 2016–2022 M&A transactions and trends in enterprise technological innovation.

Compared with existing research, this study contributes to the existing literature in four main aspects. First, it examines the impact of M&A on technological innovation and green technological innovation in the new energy industry, specifically focusing on the lithium battery sector. Existing research has primarily investigated the effects of M&A activities on technological innovation in sectors such as the agriculture industry, manufacturing industry, and high-tech industry. Due to the rapid development of the new energy sector, analyzing the influence of M&A on technological innovation in listed lithium battery companies not only enriches the literature pertaining to the effects of M&A on innovation across different industries but also provides empirical evidence that can promote high-quality development in the lithium battery industry and facilitate the green and lowcarbon transformation of the economy and society. Second, it reveals the mechanisms through which lithium battery listed companies enhance their innovation performance and green innovation performance under the M&A scenario, considering factors such as firms' knowledge absorption capabilities and intellectual property protection. Thus, the researchers' understanding of technological innovation in the new energy industry is deepened. Third, it reveals the key factors that facilitate the enhancement of technological innovation levels among lithium battery-listed companies based on different purposes of M&A in the industry. Last, from the perspective of environmental pollution control, it evaluates the impacts of changes in technological innovation and green technological innovation on pollutant emissions following M&A activities.

This paper is structured as follows: Section 2 presents a literature review, upon which hypotheses pertaining to the impact of M&A on technological innovation and green technological innovation in the new energy industry are proposed. Section 3 describes the

utilized empirical data and methodology. Section 4 presents and discusses the main results of the econometric model. Finally, Section 5 provides targeted policy recommendations for new energy companies based on a summary of the empirical research findings.

2. Literature Review and Hypothesizes Proposed

2.1. *Literature Review*

In the existing research on the technological innovation of enterprises, many studies begin their analysis with factors such as environmental regulation [3,4], stakeholder pressure [5,6], enterprise resource capability and executive characteristics [7–10], and corporate M&A behavior [11,12], which represent the driving factors for enterprise technological innovation activities. The relationship between M&A and technological innovation activities has attracted immense research attention; however, the research conclusions remain controversial. On the one hand, some researchers propose that M&A can provide enterprises with complementary innovation resources, expand the knowledge stock of enterprises, and reduce the risk and cost of innovation [13]. Meanwhile, M&A integration is conducive to expanding the scale of innovation and realizing economies of scale and scope in R&D [14], which enhances the innovation level. On the other hand, some studies have observed that corporate M&A weakens competition within the industry and between upstream and downstream sectors, thereby reducing R&D investment and hindering technological innovation [11]; meanwhile, M&A may create a scenario in which companies accumulate more debts, and the increased debt will compel companies to reduce R&D investment [15,16].

Therefore, some studies further analyze the impact of M&A on corporate innovation performance based on the M&A exhibited by different industries. First, some scholars divide M&A into horizontal mergers, vertical mergers, and hybrid mergers, and they observe that horizontal mergers and vertical mergers facilitate technological innovation among listed agricultural companies, whereas hybrid mergers will reduce their R&D investment [17]. Some studies divide M&A into technical M&A and non-technical M&A, and they observe that technological M&A can directly reduce the repetitive R&D investment pertaining to the closed innovation of listed manufacturing companies, overcome the rigidity of inherent innovation capabilities, and enhance enterprises' breakthrough innovation capabilities [18–20], whereas non-technological M&A exert a negative impact on innovation in manufacturing listed companies [21]. Meanwhile, some scholars have further investigated the impact of M&A in the equipment manufacturing industry on corporate technological innovation, and they observed that for the equipment manufacturing industry, M&A has promoted a substantial increase in the R&D capital and R&D level of listed companies [22,23]. In addition, some studies have analyzed the impact of M&A pertaining to high-tech industries on corporate technological innovation and observed that although there is a significant "inverted U-shaped" relationship between the scale of M&A and innovation performance, and that the absorptive capacity can move the inflection point of the "inverted U-shaped" to the right.

The aforementioned studies have analyzed the relationship between M&A and corporate technological innovation, and they have examined the heterogeneous impact of M&A on the technological innovation of listed companies in agriculture, manufacturing, and equipment manufacturing. However, due to the increasing demand for new energy vehicles under the current scenario, the lithium battery industry is rapidly developing, capital competition is fierce, and M&A are frequent [23,24]; however, few scholars have investigated the impact of M&A on the technological innovation of the lithium battery industry and the impact of M&A on the technological innovation of lithium battery enterprises. Because this relationship remains unclear, relevant research is urgently required. For the existing research on the lithium battery industry M&A, only a limited number of scholars have investigated the characteristics and driving factors of M&A. Monge et al. (2018) considered the M&A of US oil and gas companies in the lithium battery industry. With regard to the lithium battery industry, natural gas companies are dedicating an increasing amount of research attention to M&A [25]. Monge et al. (2020) analyzed the time series characteristics of M&A activities in the lithium battery industry [26]. Monge et al. (2021) utilized time series data to examine the impact of lithium ore resource prices on M&A behavior in the lithium battery industry, and they observed that the two are highly correlated [27].

In summary, the existing research has extensively discussed the causal relationship between M&A and corporate technological innovation, which provides an effective reference for this study; however, the following points should be enhanced. First, in the existing research on M&A and corporate technological innovation, only the M&A of enterprises associated with industries such as agriculture, manufacturing, and equipment manufacturing are considered. Therefore, further research on the M&A of the lithium battery industry is imperative. Second, in the existing research on the M&A behavior of the lithium battery industry, several studies are limited to examining the characteristics and driving factors of the M&A behavior. To overcome the limitation pertaining to lithium resource supply, the technological innovation capabilities of enterprises in the lithium battery industry should be enhanced; therefore, further research is imperative. The impact of M&A on the technological innovation of lithium battery enterprises should be investigated. Finally, studies on the relationship between M&A and corporate technological innovation utilize either the facilitation theory or suppression theory, two opposite perspectives, and the impact of M&A on corporate technological innovation at different industry levels (e.g., agriculture, manufacturing, and equipment manufacturing) has not been determined. The unified conclusion is as follows: there is an urgent need to further investigate the impact of M&A on the technological innovation of lithium battery enterprises; thus, further empirical evidence for the relationship between the two can be provided.

2.2. Hypothesis Proposed

Based on the objective of corporate transaction M&A, the strategy is divided into the following categories: technical M&A and non-technical M&A [21]; According to the direction of corporate transaction M&A, the strategy can be divided into vertical mergers, horizontal mergers, and hybrid mergers [17]. On the one hand, from the perspective of M&A in the lithium battery industry, the trend pertaining to vertical M&A of battery companies and battery material companies is apparent, and they mainly extend upstream to ensure the supply of raw materials and cost control. On the other hand, in regard to the objectives of M&A in the lithium battery industry, technological M&A is gradually becoming a new trend, with a focus on new materials and technologies (e.g., silicon-based anodes and solid-state batteries), which have attracted significant attention from acquiring capital [2]. Therefore, China's lithium battery industry mainly exhibits vertical M&A. In the analysis, it is necessary to focus on the vertical M&A of the lithium battery industry and technology mergers and to respectively explain the impact of the two on the technological innovation of enterprises.

The technological innovation concept originated from Joseph Schumpeter's theory of innovation theory, which proposes that "innovation" is an economic process of rearranging and combining original factors of production into new production methods to improve efficiency and reduce costs. The theory entails introducing new combinations of production factors and conditions into the production system; thus, a new production function with the aim of attaining potential profits can be established. Scholars subsequently categorized innovation into product innovation, process innovation, market innovation, resource allocation innovation, and organizational management innovation based on the preceding discourse. The technological innovation theory, which emerged as a distinct field from Schumpeter's innovation theory beginning in the 1950s, is defined as an integrated process where entrepreneurs seize market opportunities for potential profitability; restructure production conditions and factors; continuously develop and introduce new products, processes, and technologies; and aim for market recognition while pursuing economic benefits. Based on the object of technological innovation, it can be classified into product innovation and process innovation. Product innovation lacks a strict and unified definition, and the OECD (Organization for Economic Co-operation and Development) defines it as

the technical changes of a product that provide novel or enhanced services to product users. Process innovation, also referred to as process technological change, encompasses innovations in production techniques, including new processes, equipment, and organizational management methods. In the following analysis, which considers new energy companies, the technological innovation effects of implementing M&A strategies will be primarily examined from the product innovation and process innovation perspectives.

First, from the perspective of vertical M&A in the lithium battery industry, lithium battery companies can reduce the production cost and enhance competitiveness by acquiring upstream lithium mining resources, thereby gaining a competitive advantage, accelerating capital accumulation, increasing investment in research and development, and promoting technological innovation within the company. Additionally, vertical integration through upstream M&A facilitates the substitution of internal mechanisms for market transaction mechanisms, reducing transaction costs associated with knowledge exchange. Moreover, the aforementioned strategy enables the utilization of coordination advantages within the organization, which promotes knowledge transfer, recombines knowledge components, and ultimately enhances the company's technological innovation capabilities [28,29].

Second, from the perspective of technological M&A in the lithium battery industry, such strategies provide external research and development resources to the company, which enables synergistic innovation effects through the collaboration of external and internal research and development, thereby enhancing R&D efficiency [12,14]. External R&D effectively complements internal R&D. Through technology M&A, companies can introduce cutting-edge technologies from external competitors and products, convert external knowledge into internal knowledge, and transform external technologies into internal technologies, which reduces the internal R&D risks associated with innovation investment and enhances the company's internal R&D capabilities. On the other hand, internal R&D capabilities crucially enhance the efficiency induced by incorporating external R&D. Internal R&D ensures that the acquired technology and knowledge are absorbed internally, which leads to technological innovation, new products, and new technologies. The aforementioned activity enhances the utilization efficiency of technological M&A. Based on the preceding analysis, this study proposes the following hypothesis:

Hypothesis 1. *Lithium battery industry M&A crucially enhance the level of technological innovation of enterprises.*

Although green technological innovation is encompassed within technological innovation, it is distinguished by its focus on addressing environmental issues and achieving specific sustainable development-oriented environmental goals. Green technological innovation encompasses the design of green products and process innovations in enterprises, including aspects related to energy efficiency, pollution prevention, and waste recycling, and it also entails supportive organizational management and innovative implementation. In comparison to technological innovation, green technological innovation often exhibits characteristics such as high investment, long cycles, and significant risks. New energy companies, following M&A, may lack strong intentions for green innovation. Furthermore, the issue of incomplete and inadequate intellectual property protection in China still persists, and green technological innovation exhibits externalities, allowing competitors to utilize green innovation technologies through imitation or other means at no cost. Consequently, new energy companies lack incentives for green innovation after implementing M&A strategies. Therefore, due to the high investment, long cycles, and significant risks associated with green technological innovation, coupled with issues related to incomplete intellectual property protection in China, M&A may not significantly enhance the level of green technological innovation in new energy companies. Based on the preceding analysis, this study proposes the following hypothesis:

Hypothesis 2. *The implementation of acquisition strategies by new energy companies exerts an insignificant impact on green technological innovation.*

The knowledge absorption capacity of new energy companies is an inherent capability for identifying, acquiring, and transforming external innovation resources within the M&A context. While the implementation of acquisition strategies enables access to external innovation resources, these resources cannot be directly translated into technological innovation. Instead, new energy companies should absorb and integrate internal and external knowledge to leverage it for technological innovation [30]. Therefore, the knowledge absorption capacity of new energy companies serves as a crucial moderating mechanism that significantly impacts the relationship between M&A and innovation. The innovation effects of M&A are reliant on the recombination of knowledge components. The stronger the knowledge absorption capacity of new energy companies, the faster they can integrate external innovation knowledge, which results in higher knowledge reconfiguration efficiency. This phenomenon enables more effective utilization of the various innovative resources obtained through M&A, which facilitates the internalization of external technological expertise and strengthens the positive effects of implementing M&A on technological innovation. Thus, companies with a higher knowledge absorption capacity can rapidly assimilate and leverage external heterogeneous technologies, which accelerates the pace of technology transfer and facilitates enterprise innovation. Additionally, from the perspective of green technological innovation in new energy companies, organizations with a stronger absorption capacity tend to exhibit more robust internal mechanisms, which are advantageous in mitigating risks associated with green technological innovation. Furthermore, these companies often embrace more modernized management concepts, which contribute to an increased willingness for green innovation in new energy companies. Based on the aforementioned analysis, this study proposes the following hypothesis:

Hypothesis 3. The stronger the knowledge absorption capacity of new energy companies, the more beneficial the implementation of acquisition strategies that promote technological innovation and enhance the level of green technological innovation in new energy companies.

The external nature of technological innovation implies that the strength of intellectual property protection at the regional or national level in which a company operates will significantly impact the willingness and propensity for technological innovation. For new energy companies, the positive impact of the implementation of acquisition strategies on technological innovation is influenced by the level of intellectual property protection in their respective regions or countries. Strengthening intellectual property protection can somewhat restrict imitation by peers or competitors, which mitigates potential risks associated with innovation and increases the value of research and development outcomes. Consequently, the innovation willingness of new energy companies following M&A is enhanced, thereby promoting technological innovation. Additionally, strengthening intellectual property protection can reduce the externalities of green technological innovation, which safeguards it and boosts the willingness for green innovation in new energy companies post-acquisition. Furthermore, intellectual property protection can enhance the market competitiveness of energy companies, bolstering their ability to resist risks associated with green innovation. Post-acquisition, new energy companies are more willing to integrate resources and strengthen green technological innovation, which yields more significant positive effects on green technological innovation in new energy companies. Based on the preceding analysis, this study proposes the following hypothesis:

Hypothesis 4. Strengthening intellectual property protection will facilitate the implementation of acquisition strategies that promote technological innovation and enhance the level of green innovation in new energy companies.

The M&A activities of new energy companies can generally be divided into two categories: technology-based mergers and non-technology-based mergers. On the one hand, non-technology-based mergers in the lithium battery industry mainly entail vertical mergers. Vertical mergers refer to the acquisition of companies that are upstream or downstream of the acquiring company, with a direct relationship in regard to production processes or business operations. Vertical mergers can reduce search costs, negotiation costs, and transaction costs pertaining to the merger process, which mitigates opportunistic behavior and resulting breach costs, and ultimately enhances overall performance and achieves monopolistic profits. However, the motivation to implement such mergers rarely stems from the innovative resources of the target company, as they may not transfer suitable technological resources or R&D talents to the acquiring company. On the other hand, new energy companies achieve technological innovation through two modes of technologybased mergers, namely related mergers and unrelated mergers. In the lithium battery industry, related mergers, where the acquiring and target companies belong to the same industry and possess similar technological domains, are more prevalent. Using technologybased mergers, the acquiring company can complement its existing strengths; integrate, refine, and explore the merger resources; advance along the existing technological path; achieve technological breakthroughs; and benefit from the advantages of low information asymmetry, low merger risks, short learning time, high learning efficiency, and strong synergy effects. Moreover, mergers can induce suitable R&D resources and talents to the acquiring company. Talent is the primary driving force and decisive factor for green technological innovation. The entire process pertaining to R&D activities, the application of technological innovation outcomes, and industrialization are heavily reliant on the strong support of high-level human capital. The implementation of technology-based mergers by new energy companies can accumulate human capital for green technological innovation, reduce green innovation risks, and enhance green innovation efficiency. Based on the preceding analysis, this study proposes the following hypothesis:

Hypothesis 5. Compared with non-technology-based mergers, the implementation of technologybased merger strategies not only enhances technological innovation levels but also improves the level of green innovation in new energy companies.

3. Research Framework

3.1. Empirical Model Setting

The decision-making process for technological innovation in lithium battery companies is highly complex and influenced by various factors. The double-difference model, which strictly delineates treatment and control groups, utilizes the method of differencing before and after interventions and simultaneously differencing treatment and control groups; thus, the influence of other time-varying factors on technological innovation is effectively eliminated. Under certain conditions, the model enables the estimation pertaining to the net effect of M&A on technological innovation in new energy companies. Moreover, considering the simultaneous impact of inherent firm characteristics and macroeconomic conditions as confounding factors on both M&A and technological innovation decisions in the lithium battery industry, this study incorporates firm-fixed effects, year-fixed effects, and firm-level control variables into the double-difference model represented by Equation (1). By utilizing panel data obtained from the lithium battery industry, the study aims to assess the impact of M&A on technological innovation in lithium battery companies.

$$innovation_{i,t} = \alpha_0 + \alpha_1 merge_{i,t} + \beta X_{i,t} + \mu_i + \eta_t + \varepsilon_{i,t}$$
(1)

In Equation (1), *innovation*_{*i*,*t*} represents the technological innovation level of company *i* in year *t*; *merge*_{*i*,*t*} represents the M&A behavior of company *i* in year *t*, as a dummy variable; α_0 represents the constant term; α_1 represents the estimated coefficient of the key explanatory variable *merge*_{*i*,*t*}, and $\varepsilon_{i,t}$ represents the error term. In the empirical analysis, the

estimated coefficient α_1 of the key explanatory variable $merge_{i,t}$. represents the main factor. A significant positive α_1 indicates that M&A are beneficial for enhancing the technological innovation level of new energy companies, whereas a negative α_1 indicates the opposite (i.e., M&A are not conducive to enhancing the technological innovation level of new energy companies).

The technological innovation of new energy enterprises is not only influenced by M&A strategies but also by other factors. Failure to control for other confounding factors would result in the correlation between residuals and the key explanatory variable, which leads to biased coefficient estimates.

First, to address this issue, we first consider the enterprise-level characteristics (i.e., the size, operational status, and market value of new energy companies), which simultaneously affect both their M&A decisions and innovation efficiency. By including these characteristics as control variables, herein denoted as $X_{i,t}$, endogeneity issues can be alleviated. Therefore, we further incorporate a series of control variables $X_{i,t}$ in the empirical model, with β representing the estimated coefficients of the control variables.

Second, specific factors related to new energy enterprises (i.e., corporate culture, strategic objectives, management expertise, organizational structure, and resource endowment), which cannot be quantified, may introduce biases into the estimation results. Because obtaining data on these characteristics is not feasible, we assume that they do not change over time, and we control for them by adding firm fixed effects denoted by μ_i in the empirical model.

Finally, the M&A decisions and technological innovation of new energy enterprises are also influenced by factors such as macroeconomic conditions and the COVID-19 pandemic. In the empirical model, we assume that the impact of macroeconomic conditions and the COVID-19 pandemic on new energy enterprises is homogeneous and can be controlled by introducing year-fixed effects denoted by η_i .

Figure 2 represents the research design flowchart.



Figure 2. Research Design Flowchart.

3.2. Variable Definition

(1) Sample selection

Considering the specific influence of factors such as the 2008 financial crisis and economic collapse on the development of the new energy industry, this study selects publicly listed companies in the lithium battery industry from 2012 to 2022 as the research subjects: (1) According to the main business field of companies recorded in the CSMAR listed company database, 34 listed companies whose main business covers lithium resources are screened. (2) Based on the stock code fields of 34 lithium resources listed companies, it is matched with the CSMAR M&A database, and after excluding the failed M&A, a total of 257 M&A of 34 lithium-resource listed companies were obtained during the sample observation period. According to the merger and reorganization ID defined by the CSMAR database, the final sample mainly includes three types of M&A events, namely: asset M&A (S3001), absorption mergers (S3004), and equity transfers (S3008). (3) Multiple M&A by listed companies in the same year. Among the 34 lithium resources listed companies, 24 listed companies affected M&A during the sample observation period, and 10 listed companies had never conducted M&A. The data pertaining to M&A, financial indicators, and corporate governance of 34 lithium resources listed companies was obtained from the CSMAR database.

(2) Explained variable

First, enterprise technological innovation includes green technological innovation, which is crucial for the green and low-carbon transformation of lithium battery companies. Therefore, this study selects both technological innovation and green technological innovation as the dependent variables. Second, to accurately measure enterprise technological innovation and green technological innovation, it is necessary to analyze the number of patents held by new energy companies, as well as examine the novelty, value, and significance of these patents to the new energy industry. Because the novelty and value of patents are difficult to measure directly, this study adopts measurement methods utilized by scholars such as Hagedoorn and Cloodt, as well as Issah [31,32]. Patents are recognized as generated knowledge [33] and serve as indicators of successful R&D efforts by companies [34]. Therefore, this study primarily measures technological innovation in the new energy industry from the innovation output perspective. Finally, the number of patents includes both patent applications and patent grants. Considering the longer timeframe for patent grants in China, by selecting the number of patent applications and green patent applications, researchers can examine the impact of M&A on technological innovation in the lithium battery industry in a more timely and effective manner [35,36]. The data on the number of patent applications and green patent applications for the 34 listed lithium resource companies are obtained from the CNRDS database.

(3) Core explanatory variable

During the sample observation period, 10 listed companies had never been subjected to M&A, and 24 listed companies had experienced multiple M&A. The multi-period, double-difference model can be utilized to estimate the relationship between M&A and the technological innovation of Chinese lithium industry enterprises. Utilize 24 listed companies with M&A as the treatment group, and set 10 listed companies without M&A as the control group. Based on the year when the 18 listed companies in the lithium battery industry first implemented M&A during the sample observation period, the merger value is 1 after the year of the first merger, and the merger value is 0 for the subsequent mergers. In addition, because the 18 lithium resource-listed companies with M&A implemented only M&A strategies in certain years, assigning values based on the year of the first M&A may lead to crucial estimation errors. In the robustness test, this study re-measures the core explanatory variables based on whether listed companies in the lithium battery industry implement M&A year by year. If enterprise i implements M&A in year t, the value is set at 1; otherwise, the value is set at 0.

(4) Other variables

This study adds the following control variables to the empirical model to control the impact of enterprises' observable characteristics on technological innovation. (1) Size of listed companies (*size*): the logarithmic value pertaining to the company's total assets; (2) performance of listed companies (*roa*), which is determined by the return on assets; (3) financial leverage of listed companies (*lev*), which is determined by the asset–liability ratio; (4) listed companies' company growth (*growth*), which is determined by the main business income growth rate; (5) and enterprise value (*tobin*), which is expressed as (stock market value + net debt)/total assets.

Additionally, the following variables should be measured in the empirical analysis: (1) Knowledge absorption capacity of lithium battery companies (*absorb*), referring to the measurement methods developed by Cohen and Levintha [13], which utilize the ratio pertaining to the R&D investment and patent applications of listed companies in the lithium industry to measure enterprises' absorptive capacity; this indicator measures the R&D investment consumption of a single patent application, and the smaller the ratio, the stronger the corporate absorptive capacity. (2) Intellectual property rights protection (IPRP), which is based on Lv et al.'s measurement of intellectual property rights protection [37], utilizes the quasi-natural experiment pertaining to the Chinese government's establishment of the Intellectual Property Demonstration City policy; after the policy implementation, a dummy variable is assigned a value of 1, whereas, in all other scenarios, the dummy variable is assigned a value of 0. (3) Technology M&A (technology) is based on the field "whether it involves intellectual property M&A" in the CSMAR merger and reorganization database; technology is assigned a value of 1, whereas it is assigned a value of 0 in other scenarios. (4) Listed company wastewater discharge (lnWater) represents the natural logarithm of wastewater discharge contained in the annual reports of listed companies. (5) Listed company sulfur dioxide emissions $(lnSO_2)$ represent the natural logarithm of sulfur dioxide emissions contained in the annual reports of listed companies. The measurement is based on the wastewater discharge contained in the annual reports of listed companies, and all continuous variables are subject to truncation at the 1st percentile.

The descriptive statistics of all the aforementioned variables are depicted in Table 1. There is a significant disparity in the level of technological innovation among new energy companies. On the one hand, in regard to technological innovation, the mean number of patent applications for listed companies in the lithium battery industry is 33.0025, with a maximum of 1051, a minimum of 0, and a standard deviation of 116.6709. On the other hand, with regard to green technology innovation, the mean number of green patent applications for listed companies in the lithium battery industry is 7.9237, with a maximum of 232, a minimum of 0, and a standard deviation of 30.2514. This observation indicates a significant disparity and high volatility in the technological innovation and green technology innovation within the lithium battery industry among listed companies. Therefore, in the empirical model, this study utilizes the logarithm pertaining to the sum of 1 added to the number of patent applications and green patent applications for listed study utilizes the logarithm pertaining to the sum of 1 added to the number of patent applications and green patent applications for listed companies. Therefore, in the lithium battery industry. Moreover, the mean value of the core explanatory variable is 0.2735, which indicates that 27.35% of the listed company samples in the research sample implemented M&A strategies.

Variables	Obs	Mean	Std.dev	Min	Max
patent	315	33.0025	116.6709	0.0000	1051
In <i>patent</i>	315	2.0212	1.5731	0.0000	6.9565
gpatent	315	7.9237	30.2514	0.0000	232
lngpatent	315	0.5425	1.2555	0.0000	5.4467
merge	315	0.2735	0.4464	0.0000	1.0000

 Table 1. Descriptive statistics.

Variables	Obs	Mean	Std.dev	Min	Max
size	315	21.6542	1.1425	17.9652	22.9562
roa	315	0.0552	0.0541	0.0072	0.2625
lev	315	3.1475	2.1426	1.5865	13.2695
growth	315	0.3256	0.3526	-0.1028	2.3175
tobin	315	2.2651	1.0213	1.2517	8.4562
absorb	315	15.1617	1.0340	12.1837	18.0357
IPRP	315	0.2875	0.3845	0	1
technology	315	0.2941	0.4561	0	1
ln <i>Water</i>	312	3.2445	2.3527	0	9.3984
$lnSO_2$	308	4.5043	1.9146	0	12.2294

Table 1. Cont.

4. Empirical Analysis

4.1. Parallel Trend Test

Before evaluating the impact of lithium battery industry M&A on corporate technological innovation, it is necessary to verify whether the treatment group and the control group satisfy the parallel trend assumption. Based on the year when the listed companies in the lithium battery industry first implemented M&A as the base period, construct dummy variables merge_6, merge_5, merge_4, merge_3, merge_2, and merge_1 to respectively represent 5 years, 4 years, 3 years, 2 years, and 1 year before the implementation of M&A; and construct dummy variables merge1, merge2, merge3, merge4, merge5, and merge6 to represent 1 year, 2 years, 3 years, 4 years, 5 years, and 6 years after the implementation of the merger, respectively. The core explanatory variable merge, contained in Model (1), is replaced by the aforementioned dummy variables, and Figure 3 depicts the results of the parallel trend test. Before the listed companies in the lithium battery industry implement the M&A strategy, the estimated coefficients of the dummy variables are insignificant, which indicates that there is no significant difference in technological innovation between the listed lithium battery-industry companies in the treatment group and those in the control group; thus, the parallel trend assumption is satisfied, and the multi-period difference-in-differences approach can be utilized to assess the impact of M&A on lithium battery listed companies. Meanwhile, after the implementation of the M&A strategy, the estimated coefficient of the dummy variable is significantly positive at the 10% level, which initially indicates that M&A can enhance the technological innovation capabilities of listed lithium battery companies.



Figure 3. Parallel trend test.

4.2. Basic Regression

The parallel trend test indicates that there is no significant difference in the level of technological innovation between the treatment group and the control group before the implementation of M&A and that the multi-period double difference model (1) can be estimated. Table 2 illustrates the estimation results. All empirical models include firm-fixed effects and year-fixed effects. The dependent variables in columns (1) and (2) are the number of patent applications. Regardless of whether control variables are included or not, the estimated coefficient of the core explanatory variable is approximately 0.015, which is significant at a 5% level or lower, and this value indicates a positive effect. In other words, compared with lithium battery industry-listed companies that have not implemented M&A, those that have implemented M&A in the lithium battery industry exhibit a significant increase of approximately 1.5% in the number of post-merger patent applications. This observation verifies Hypothesis 1. The empirical results indicate that approximately 2% of the patent application quantity in the lithium battery industry can be attributed to the M&A activities of new energy companies. Because M&A in the lithium battery industry significantly enhances the technological innovation level of listed companies, this observation bears significant economic implications. In columns (3) and (4), the dependent variable is the number of green patent applications. The estimated coefficients of the core explanatory variable are positive and significant when control variables are not included; however, they become insignificant after controlling for other factors. This observation indicates that the positive effect of M&A on the green technology innovation capacity of the listed lithium battery-industry is not statistically significant, thereby confirming Hypothesis 2, which may be attributed to the characteristics of green technology innovation (e.g., long cycles, high investment, and high risks). Additionally, the incomplete intellectual property protection system in China and the low willingness of new energy companies to engage in green technology innovation after M&A could contribute to this result.

	(1)	(2)	(3)	(4)
	patent	patent	gpatent	gpatent
merge	0.0152 ***	0.0147 **	0.0043 *	0.0025
, i i i i i i i i i i i i i i i i i i i	(0.0032)	(0.0073)	(0.0025)	(0.0031)
size		-0.7952 ***		-0.3414
		(0.2552)		(0.2854)
roa		6.6712		1.5754
		(4.3785)		(5.2754)
lev		0.0051		-0.0741
		(0.3758)		(0.2785)
growth		0.0092		-0.3932
		(0.3452)		(0.2946)
tobin		0.3120		0.5886 **
		(0.4125)		(0.2586)
constant	2.6852 ***	19.3586 ***	1.2586 ***	7.6875
	(0.1400)	(5.4856)	(0.1305)	(6.2385)
corporate fix effect	Yes	Yes	Yes	Yes
year fix effect	Yes	Yes	Yes	Yes
N	315	315	315	315
R ²	0.7836	0.9458	0.7836	0.9458

Table 2. Basic regression.

*, **, *** Significantly at the 10%, 5%, and 1% levels, respectively. The robust standard error of heteroscedasticity is reported in parentheses.

Examining the dynamic impact of M&A conducted by new energy companies on technological innovation is a crucial research focus herein. The examination not only clarifies the sustainability pertaining to the innovation effects of merger strategies but also provides further evidence for Hypotheses 1 and 2. In the empirical analysis, the core explanatory variable is lagged by one period, two periods, three periods, and four periods; thus, Model (1) is re-estimated, and the sustained impact of new energy company M&A on technological innovation is examined. Table 3 illustrates the estimation results with technological innovation as the dependent variable. The results in columns (1), (2), (3), and (4) correspond to the estimation results, with the core explanatory variable lagged by one period, two periods, three periods, and four periods, respectively. It can be observed that the lagged coefficients of the core explanatory variable are significant, at least at a 10% level, which further validates Hypothesis 1. Additionally, based on the magnitudes of the estimated coefficients, the lagged coefficients of the core explanatory variable in periods one to four are greater than the coefficient without lag, which indicates the sustainability pertaining to the impact of new energy company M&A on technological innovation. This observation could be attributed to the initial phase of merger strategies, where the integration of innovative resources is limited and exerts a smaller influence on technological innovation. However, within 2–4 years of implementing merger strategies, the deep integration of acquired innovative resources leads to a greater impact on technological innovation.

Table 3. Dynamic effects analysis: mergers and acquisitions and technological innovation in the new energy sector.

	(1)	(2)	(3)	(4)
	patent	patent	patent	patent
L1.merge	0.0163 **			
L2.merge	(0.0077)	0.0282 ***		
L3.merge		(0.0000)	0.0268 **	
L4.merge			(0.0003)	0.0232 *
constant	3.3385 *** (0.1200)	4.5861 *** (0.5286)	4.8675 *** (0.8564)	2.1358 ***
control variables	Yes	Yes	Yes	Yes
corporate fix effect year fix effect	Yes Yes	Yes Yes	Yes Yes	Yes Yes
N R ²	280 0.9485	245 0.8652	210 0.7963	172 0.6852

*, **, **** Significantly at the 10%, 5%, and 1% levels, respectively. The robust standard error of heteroscedasticity is reported in parentheses

Table 4 depicts the estimation results with green technological innovation as the dependent variable. The results in columns (1), (2), (3), and (4) correspond to the estimation results, with the core explanatory variable lagged by one period, two periods, three periods, and four periods, respectively. It can be observed that the lagged coefficients of the core explanatory variable are positive but not statistically significant. This observation indicates that the implementation of M&A by new energy companies does not significantly enhance their level of green technological innovation. This observation provides further validates Hypothesis 2, which indicates that the weak willingness of new energy companies to engage in green innovation after M&A is attributable to the large investment, high risks, and long cycles associated with green technological innovation, as well as the need for further improvements in China's intellectual property protection system.

	(1)	(2)	(3)	(4)
	lngpatent	lngpatent	lngpatent	lngpatent
L1.merge	0.0008 (0.0175)			
L2.merge	× ,	0.0014 (0.0027)		
L3.merge		× ,	0.0032 (0.0025)	
L4.merge			· · · ·	0.0023 (0.0458)
constant	2.4824 *** (0.1200)	2.7958 *** (0.0789)	2.7652 *** (0.3985)	1.1358 *** (0.1478)
control variables	Yes	Yes	Yes	Yes
corporate fix effect year fix effect	Yes Yes	Yes Yes	Yes Yes	Yes Yes
N R ²	278 0.9028	241 0.8635	205 0.8482	168 0.8062

Table 4. Dynamic effects analysis: mergers and acquisitions and green technological innovation in the new energy sector.

*** Significantly at the 1% level. The robust standard error of heteroscedasticity is reported in parentheses

5. Robustness Test

(1) Time placebo test

In the benchmark regression results, the impact of M&A on the technological innovation level of listed companies in the lithium battery industry is attributable to the enterprise-level intervention occasioned by the pre-merger processing group. To exclude the potentially competitive explanations, this study advances the M&A time of 24 listed lithium battery-industry companies by 3 and 5 years and re-estimates the multi-period double difference model (1) as a placebo test. Table 5 depicts the placebo test pertaining to the estimated results. Items (1) and (2) are listed as the estimated results after the M&A year was advanced by 3 years, and the estimated coefficients of the core explanatory variable merge are insignificant; items (3) and (4) are listed as the estimation results after the M&A year is advanced by 5 years, and the core explanatory variables (i.e., merge i and t estimation coefficients) are still insignificant, which indicates that for the benchmark regression results, the positive impact of M&A on the technological innovation of listed lithium battery-industry companies is not occasioned by intervention at the level of other enterprises but by the net effect of the M&A activities of listed lithium battery companies.

Table 5. Time placebo test.

	(1)	(3)	(2)	(4)
	patent	gpatent	patent	gpatent
merge	-0.0003	0.0072	0.0027	0.0057
C	(0.0039)	(0.0041)	(0.0045)	(0.0078)
size	0.0035	0.5468	0.0004	0.5719
	(0.7543)	(0.6583)	(0.8054)	(0.8759)
roa	9.0135 **	6.0709	10.5331 **	5.0791
	(4.1298)	(5.3857)	(4.1298)	(5.4805)
lev	0.4013 **	0.1119	0.4134 **	0.1891
	(0.1688)	(0.1845)	(0.1718)	(0.2397)
growth	0.1854	-0.1352	0.1994	-0.0750
0	(0.3594)	(0.4009)	(0.4695)	(0.3736)
tobin	-0.0167	0. 5854	-0.0287	0.3484
	(0.3225)	(0.3598)	(0.3762)	(0.3195)
constant	2.7463 ***	1.2493 ***	0.6850	-13.3128
	(0.1268)	(0.1338)	(18.3011)	(19.6353)
corporate fix effect	Yes	Yes	Yes	Yes
year fix effect	Yes	Yes	Yes	Yes
N	248	248	190	190
\mathbb{R}^2	0.8384	0.8296	0.9619	0.9258

, * Significantly at the 5%, 1% levels, respectively. The robust standard error of heteroscedasticity is reported in parentheses.

(2) Replace the core explanatory variables

In the preceding empirical analysis, this study constructs a dummy variable as the core explanatory variable directly based on the year when the listed company first implemented M&A. However, listed companies in the lithium battery industry did not implement M&A in some years, and the positive effect in the benchmark regression results is attributable to variable measurement bias. To exclude the aforementioned competitive explanations, this study measures the core explanatory variables based on the year-by-year M&A of listed lithium battery-industry companies. If company i implements M&A in year t, a value of 1 is assigned; otherwise, a value of 0 is assigned. The estimated results after replacing the core explanatory variables are depicted in Table 6. The explanatory variables in columns (1) and (2) represent the technological innovations of enterprises, and the estimated coefficient of merge is significantly positive at the >10% level; (3), (4) The explanatory variable contained in column is enterprise green technology innovation, and the estimated coefficient is significantly positive, but not significant; this observation is consistent with the benchmark regression results. To a certain extent, the aforementioned competitive explanations are excluded, which provides a more reliable basis for the establishment of Hypothesis 1.

	(1)	(2)	(3)	(4)
	patent	patent	gpatent	gpatent
merge	0.0103 **	0.0133 *	0.0043	0.0011
	(0.3005)	(0.0079)	(0.0861)	(0.0069)
size		-0.8318 ***		-0.2036
		(0.2465)		(0.3128)
roa		8.2436		0.2531
		(5.0346)		(5.3603)
lev		0.0083		-0.1492
		(0.2297)		(0.2378)
growth		0.0435		-0.3078
Ũ		(0.3520)		(0.3086)
tobin		0.1381		0.7337 **
		(0.3913)		(0.3282)
constant	2.4610 ***	19.9657 ***	1.0846 ***	4.7037
	(0.0903)	(5.5333)	(0.0904)	(6.8378)
corporate fix effect	Yes	Yes	Yes	Yes
year fix effect	Yes	Yes	Yes	Yes
Ν	153	50	153	50
R ²	0.7750	0.9395	0.7306	0.8967

Table 6. Replace core explanatory variable.

*, **, **** Significantly at the 10%, 5%, and 1% levels, respectively. The robust standard error of heteroscedasticity is reported in parentheses.

(3) Robustness Testing Considering Multiple Confounding Factors

Technological innovation in enterprises is a complex process that can be influenced not only by the characteristics of the companies but also by other factors such as regional environmental regulations, technological innovation trends in different industries upstream and downstream of the new energy sector, and the marketization degree. The observed effect of technological innovation in the benchmark regression results may be influenced by the aforementioned factors. Therefore, this study progressively includes industry \times time trends, city \times time trends, regional environmental regulatory intensity, and regional marketization level as control variables in the empirical model. The measurement method for regional environmental regulatory intensity follows the approach of Yin [38], with data sourced from the "China Urban Statistical Yearbook," and the marketization level data is directly obtained from the China Marketization Index Database.

Table 7 depicts the estimation results with the aforementioned control variables progressively included. The estimation results in columns (1) and (2) correspond to the dependent variable of technological innovation, where the estimated coefficients of the core explanatory variable are significantly positive at the 10% level. The estimation results in columns (3) and (4) correspond to the dependent variable of green technological innovation, where the estimated coefficients of the core explanatory variable are not statistically significant. This observation is consistent with the benchmark regression results, which indicates that the benchmark regression results are less influenced by regional government environmental regulations, industry-specific technological development trends, and market competition. Thus, the credibility of Hypotheses 1 and 2 is enhanced.

	(1)	(2)	(3)	(4)
	patent	patent	gpatent	gpatent
merge	0.0094 *	0.0098 *	0.0015	0.0009
size	(0.0053) -0.7215 **	(0.0057) -0.7045 ***	(0.0861) -0.8569 ***	(0.0069) -0.5621 *
roa	(0.2854) 7.5284 (4.0578)	(0.1954) 7.9876 (5.8512)	(0.2438) 1.5248 (5.6875)	(0.3088) 1.5482 (4.5212)
lev	(4.9578) 0.0102 (0.3014)	0.0095	(0.0085) (0.2356)	(4.3213) -0.0109 (0.2486)
growth	0.0435	0.0435	(0.2530) -0.0435 (0.3520)	-0.3078 (0.3086)
tobin	0.1586	0.1435	0.8456 *	0.7337 **
constant	2.1258 *** (0.0903)	2.1023 *** (0.9658)	4.0825 (5.8623)	4.7037 (6.8378)
industry × time trends	Yes	Yes	Yes	Yes
$city \times time trends$	Yes	Yes	Yes	Yes
environmental regulatory intensity marketization level	No No	Yes Yes	No No	Yes Yes
corporate fix effect	Yes	Yes	Yes Vos	Yes
	100	103	100	100
N R ²	312 0.8745	310 0.8564	312 0.8912	310 0.8643

Table 7. Considering Multiple Confounding Factors.

*, **, *** Significantly at the 10%, 5%, and 1% levels, respectively. The robust standard error of heteroscedasticity is reported in parentheses.

6. Moderating Mechanism Analysis

(1) Absorptive capacity of new energy companies

The strength of knowledge absorption capability in new energy companies directly impacts the synergy of internal and external innovative resources and the rate at which knowledge costs are recombined. Additionally, stronger knowledge absorption capabilities lead to more sophisticated internal organizational structures, higher risk resilience, and greater willingness and enthusiasm for green innovation. Therefore, stronger knowledge absorption capabilities in new energy companies facilitate technological innovation and contribute to the enhancement of green technological innovation levels through the implementation of M&A. To further validate the moderating role of knowledge absorption capability in listed lithium battery-industry companies, this study constructs an interaction model as illustrated in Equation (2).

$$innovation_{i,t} = \alpha_0 + \alpha_1 merge_{i,t} \times absorb_{i,t} + \alpha_2 merge_{i,t} + \beta X_{i,t} + \mu_i + \eta_{p,t} + \varepsilon_{i,t}$$
(2)

The larger the value of variable *absorb*_{*i*,*t*}, the weaker the absorption capability of the enterprise. The estimated coefficient α_1 of the expected interaction term is significantly negative. As the absorption capability of listed companies decreases, the positive effect of M&A on technological innovation gradually weakens. In other words, enhancing the absorption capability of listed companies will lead to a stronger stimulating effect of M&A on technological innovation and green technological innovation. The estimation results of the interaction model are depicted in Table 8, where columns (1) and (2) correspond to the dependent variables of technological innovation and green technological innovation, respectively. The heteroscedasticity-robust standard errors are reported in parentheses. The estimated coefficients of the interaction terms are -0.0063 and -0.0046, and they are both significant at the 10% level, which indicates a negative effect. Additionally, in

columns (3) and (4), the estimation results cluster the standard errors at the industry level, which increases their magnitude. The coefficients remain significantly negative at the 11% level, consistent with the expected results. These empirical results demonstrate that, in the context of M&A, enhancing the knowledge absorption capability of new energy companies can simultaneously enhance the levels of technological innovation and green technological innovation, which validates Hypothesis 3.

	(1)	(2)	(3)	(4)
	patent	gpatent	patent	gpatent
Merge $ imes$ absorb	-0.0063 ***	-0.0046 *	-0.0063 **	-0.0046
-	(0.0018)	(0.0021)	(0.0021)	(0.0027)
merge	1.1369	-1.7424	1.1369	-1.7424
	(2.1619)	(6.6443)	(1.2714)	(5.0204)
absorb	-0.4508 *	-1.2425 **	-0.4508 ***	-1.2425 **
	(0.1932)	(0.4142)	(0.0824)	(0.3589)
size	-1.4007 **	0.6025	-1.4007 **	0.6025
	(0.5039)	(1.3077)	(0.3838)	(0.9349)
roa	-0.2253	-14.7900	-0.2253	-14.7900 *
	(4.0680)	(8.6456)	(2.8608)	(6.1018)
lev	0.5268 ***	0.0227	0.5268 ***	0.0227
	(0.1298)	(0.2591)	(0.1132)	(0.2019)
growth	-0.4861 *	-0.5946	-0.4861	-0.5946 *
	(0.2298)	(0.5293)	(0.2499)	(0.2912)
tobin	-0.3005	0.1758	-0.3005	0.1758
	(0.2583)	(0.4401)	(0.3328)	(0.1651)
constant	40.8669 ***	7.4343	40.8669 ***	7.4343
	(9.8690)	(27.6524)	(9.0488)	(17.6252)
corporate fix effect	Yes	Yes	Yes	Yes
year fix effect	Yes	Yes	Yes	Yes
N	315	315	315	315
\mathbb{R}^2	0.9731	0.9193	0.9731	0.9193

Table 8. Analysis of moderating mechanisms: absorptive capacity.

*, **, *** Significantly at the 10%, 5%, and 1% levels, respectively. The robust standard error of heteroscedasticity is reported in parentheses in columns (1) and (2), and the standard error of clustering residuals to the enterprise level is reported in parentheses in columns (3) and (4).

(2) The level of intellectual property protection

The level of intellectual property protection is a critical factor that affects the innovation willingness of listed lithium battery-industry companies. The lack of intellectual property protection can significantly restrain green technological innovation in new energy enterprises. By strengthening the level of intellectual property protection, new energy companies are more likely to actively integrate the knowledge and technologies related to production and environmental protection from both the acquiring and acquired parties after implementing M&A strategies. This strategy will simultaneously enhance the levels of general technological innovation and green technological innovation. To further validate the moderating effect of intellectual property protection on green technological innovation in new energy companies, this study constructs an interaction model represented by Equation (3).

$$innovation_{i,t} = \alpha_0 + \alpha_1 merge_{i,t} \times IPRP_{i,t} + \alpha_2 merge_{i,t} + \beta X_{i,t} + \mu_i + \eta_{p,t} + \varepsilon_{i,t}$$
(3)

 $IPRP_{i,t}$ represents the level of intellectual property protection in the city where the new energy company is located. Considering Lv et al.'s measurement of intellectual property protection at the city level in China [37], and based on the Chinese government's pilot policy (i.e., to establish intellectual property demonstration cities), this study measures the intellectual property protection environment faced by enterprises. The larger the value

of variable *IPRP*_{*i*,*t*}, the stronger the intellectual property protection; therefore, this study expects the estimated coefficient α_1 of the interaction term to be significantly positive. In other words, the stronger the intellectual property protection, the stronger the positive effect of implementing M&A strategies on technological innovation in new energy companies.

The estimation results of the interaction model are illustrated in Table 9, with the heteroscedasticity-robust standard errors reported in columns (1) and (2). Regardless of whether the dependent variable is a technological innovation or green technological innovation, the estimated coefficients of the interaction term are significantly positive at the 5% level, which is consistent with the expected result. Additionally, in columns (3) and (4), the estimation results cluster the standard errors at the industry level, which increases their magnitude and exhibits significant positive effects at the 10% level. This observation indicates that intellectual property protection is a crucial moderating variable. As the intellectual property protection environment improves for listed companies in the new energy industry, M&A can simultaneously promote the levels of technological innovation and green technological innovation. Thus, Hypothesis 3 is validated.

	(1)	(2)	(3)	(4)
	lnpatent	lngpatent	Inpatent	Ingpatent
merge $ imes$ IPRP	0.0021 **	0.0015 **	0.0021 *	0.0015 *
, , , , , , , , , , , , , , , , , , ,	(0.0009)	(0.0007)	(0.0011)	(0.0009)
merge	1.0018	-1.0025	1.0018	-1.0025
-	(1.5269)	(5.5678)	(2.0031)	(5.8478)
IPRP	-0.3807 **	-1.3842 ***	-0.3807 **	-1.3842 **
	(0.1892)	(0.4052)	(0.1658)	(0.5379)
size	-1.3584 **	0.6025	-1.3584 ***	0.6025
	(0.6235)	(1.5368)	(0.4528)	(0.8658)
roa	-0.3658	-12.35	-0.3658	-12.35
	(3.5869)	(8.6847)	(2.3896)	(7.3585)
lev	0.6258 ***	0.0385	0.6258 ***	0.0385
	(0.1486)	(0.2756)	(0.1052)	(0.3052)
growth	-0.3754 *	-0.5012	-0.3754	-0.5012
-	(0.2186)	(0.5463)	(0.4685)	(0.2146)
tobin	-0.2975	0.1518	-0.2975	0.1518
	(0.2485)	(0.1845)	(0.3328)	(0.1651)
constant	41.8465 ***	7.2468	41.8465 ***	7.2468
	(9.3257)	(28.5645)	(9.2578)	(18.2565)
corporate fix effect	Yes	Yes	Yes	Yes
year fix effect	Yes	Yes	Yes	Yes
N	315	315	315	315
R ²	0.9731	0.9193	0.9731	0.9193

Table 9. Analysis of moderating mechanisms: intellectual property protection.

*, **, **** Significantly at the 10%, 5%, and 1% levels, respectively. The robust standard error of heteroscedasticity is reported in parentheses.

7. Heterogeneity Analysis

When listed companies in the lithium battery industry implement M&A strategies, they can opt for either technical mergers or non-technical M&A. Non-technical M&A mainly extend upstream through vertical M&A to ensure the supply of raw materials and reduce costs. To analyze the impact of M&A in the lithium battery industry on technological innovation, it is necessary to further elaborate on which type of M&A crucially facilitates the improvement of technological innovation levels of enterprises. Herein, based on whether the M&A activities of listed companies in the lithium battery industry entailed intellectual property M&A during the sample observation period, the 34 listed companies were divided into technology M&A sample groups and other M&A sample groups and re-estimated using the empirical model (1) in the sub-sample, and the results are depicted in Table 10. Items (1) and (2) represent the estimated results pertaining to the technology

M&A sub-sample. The explained variables are enterprise technology innovation and green technology innovation. The estimated coefficients of the core explanatory variable merge are 0.0258 and 0.0079, at least at the 10% level. The lower value is significant, which indicates that technology M&A has significantly enhanced the technological innovation capabilities of listed lithium battery-industry companies. Columns (3) and (4) are the estimated results of the non-technical M&A sample group, and the explained variables are corporate technology innovation and green technology innovation. Compared with the estimated results in columns (1) and (2), the core explanatory variables, the estimated coefficient of a merge is smaller, and the statistical significance is significantly reduced, which indicates that technology M&A are the main driving factor for the technological enhancement of listed companies in the lithium battery industry.

	Technology M&A		Nontechno	ology M&A
	(1) patent	(2) gpatent	(3) patent	(4) gpatent
merge	0.0258 ** (0.0131)	0.0079 * (0.0046)	0.0068 *	-0.0005 (0.0082)
size	-1.7238 (0.7353)	-4.4416 (1.7373)	-0.7275 (1.2240)	2.8596 *
roa	12.0920	33.8063	(17.1210) -1.0144 (17.1219)	-82.4669 ** (5 5301)
lev	(0.5249) (0.5497)	0.3896	0.2228 (0.9017)	-0.7422 (0.2658)
growth	-0.4846 (0.5378)	(1.2120) -1.7177 (2.6111)	-0.3046	-3.2207 (0.5724)
tobin	-0.4005	(2.0111) -0.3804 (1.1021)	(1.1737) 0.7765 (1.4441)	(0.3724) 1.2259 (0.2418)
constant	(0.6039) 37.4242 (14.6908)	(1.1031) 103.8420 (39.5838)	(1.4441) 13.3891 (24.5550)	(0.2418) -57.9323 * (8.6773)
corporate fix effect year fix effect	Yes Yes	Yes Yes	Yes Yes	Yes Yes
N R ²	95 0.9209	95 0.9982	168 0.6700	168 0.9999

Table 10. Heterogeneity analysis.

*, ** Significantly at the 10%, 5% levels, respectively. The robust standard error of heteroscedasticity is reported in parentheses.

8. Further Research

For new companies, the implementation of M&A strategies can enhance technological innovation levels and promote green technological innovation under certain conditions. However, the improvement in technological innovation levels following the implementation of M&A strategies may yield unexpected implications for environmental governance. On the one hand, the enhancement of technological innovation levels can enhance production efficiency and energy efficiency in new energy enterprises, reduce fossil fuel consumption, and decrease emissions of pollutants such as carbon dioxide and sulfur dioxide, thereby generating positive effects on environmental governance. On the other hand, the improvement in energy efficiency can potentially stimulate increased energy consumption by lithium battery companies, which leads to rebound effects in energy usage. Green technological innovation mainly entails pollution control and prevention technologies, source reduction technologies, waste minimization technologies, recycling and regeneration technologies, ecological processes, green products, and purification technologies, which can reduce the emission of pollutants during the product manufacturing process. Based on the preceding analysis, this study proposes that the enhancement of technological innovation levels following the implementation of M&A strategies by new energy companies may not necessarily be beneficial for environmental governance. However, the improvement in green technological innovation levels occasioned by M&A can contribute to environmental governance.

This study constructs the interaction models represented by Equations (4) and (5) to validate the preceding analysis. In these models, $pollution_{i,t}$ represents the pollutant emissions of new energy companies, and it primarily includes sulfur dioxide and industrial wastewater emissions. The data are obtained from environmental pollutant emission disclosures contained in the annual reports of listed companies. $merge_{i,t} \times lnpatent_{i,t}$ and $merge_{i,t} \times lnpatent_{i,t}$ represent the interaction terms between M&A and technological innovation, as well as green technological innovation, respectively. The definitions of other variables remain constant. In the empirical analysis, the study primarily considers the estimated coefficients of the interaction terms $merge_{i,t} \times lnpatent_{i,t} \times lnpatent_{i,t}$. A negative coefficient indicates that the technological innovation effects resulting from M&A are detrimental to environmental governance.

The estimation results of Model (4) are presented in columns (1) and (2) of Table 11. The estimated coefficients of the interaction terms are negative; however, they are not statistically significant. This lack of significance may be occasioned by the offsetting effects pertaining to the inhibitory effect of technological innovation and the rebound effect of energy consumption, which leads to insignificant estimation coefficients for the interaction terms.

	(1)	(2)	(3)	(4)
	ln <i>Water</i>	lnSO ₂	lnWater	lnSO ₂
merge \times Inpatent	-0.0021	-0.0015		
0	(0.0019)	(0.0013)		
Inpatent	0.0038 *	0.0059 **		
-	(0.0021)	(0.0029)		
merge \times lngpatent			-0.0042 *	-0.0053 *
0 0.			(0.0025)	(0.0031)
Ingpatent			-0.0102 **	-0.0208 *
			(0.0047)	(0.0114)
merge	-1.0033	-1.0048	1.0439	1.0315
	(1.3568)	(2.8564)	(1.5746)	(2.0875)
size	-0.9487	-0.8456 *	-0.9875 **	-0.8754 *
	(0.6235)	(0.4382)	(0.4528)	(0.5234)
roa	-0.4256	-0.8546	-0.3546	-0.9784
	(0.9874)	(0.7452)	(0.3896)	(0.8645)
lev	0.5214 ***	0.4851	0.3257 ***	0.2586
	(0.1265)	(0.2012)	(0.1052)	(0.3052)
growth	-0.3856 *	-0.4965	-0.2851	-0.3019
	(0.2242)	(0.4862)	(0.2019)	(0.2859)
tobin	0.2835	0.1247	0.2975	0.1518
	(0.2485)	(0.1845)	(0.3328)	(0.1651)
constant	1.5216 ***	2.6528 ***	1.8745 ***	2.7451 ***
	(0.2145)	(0.1936)	(0.4026)	(0.2049)
corporate fix effect	Yes	Yes	Yes	Yes
year fix effect	Yes	Yes	Yes	Yes
N	301	305	298	300
\mathbb{R}^2	0.3546	0.2814	0.3546	0.2814

Table 11. Further Research.

*, **, **** Significantly at the 10%, 5%, and 1% levels, respectively. The robust standard error of heteroscedasticity is reported in parentheses.

The estimation results of the empirical model (5) are depicted in columns (3) and (4) of Table 10. The estimated coefficients of the interaction terms are -0.0042 and -0.0053, and they are significant at the 10% level. This observation indicates that the enhancement of green technological innovation levels following M&A can reduce pollutant emissions, which is consistent with the expected results.

9. Conclusions and Suggestions

Based on the background of promoting the high-quality development of the new energy industry, particularly the lithium battery industry, to achieve the "carbon peaking" goal, this study focuses on publicly listed companies in the Chinese lithium battery industry from 2012 to 2022. Using a multiple-period difference-in-differences approach, the study empirically examines the impact of M&A on technological innovation in lithium battery companies. The research findings indicate that the implementation of M&A strategies by new energy companies can enhance their technological innovation levels. However, M&A do not significantly improve the levels of green technological innovation in the listed lithium battery companies. These conclusions are robust against tests such as parallel trend tests, placebo tests, and core explanatory variable substitution tests.

Furthermore, the analysis of regulatory mechanisms reveals the following: after implementing M&A strategies, strengthening the absorptive capacity of knowledge and technology, as well as enhancing intellectual property protection, is more conducive to the improvement of technological innovation levels. Moreover, this strategy can stimulate new energy companies to actively engage in environmentally friendly green technological innovation. Using a heterogeneity analysis, it is revealed that technological M&A can simultaneously enhance both technological innovation and green technological innovation levels. Non-technological M&A, such as vertical mergers, do not significantly influence technological innovation in companies. Therefore, technological M&A are the main driving factors for enhancing technological innovation capabilities in the lithium battery industry.

Additionally, further studies reveal heterogeneity in the environmental effects of technological innovation occasioned by new energy company M&A. By implementing M&A strategies, which entails promoting green technological innovation, industrial wastewater and sulfur dioxide emissions can be effectively reduced. Therefore, this research bears clear policy implications:

(1) Lithium battery companies should develop comprehensive M&A strategies based on their own resource endowments. M&A should be perceived as a crucial lever for enhancing technological innovation capabilities. Adequate resource integration should be performed after M&A. In regard to the increasingly intense external competitive environment, lithium battery companies should not rely solely on closed innovation. They should exhaustively leverage their resource advantages and assess the technical knowledge and mineral resources of target companies. By conducting M&A, they can obtain open innovation resources and achieve high-quality development through the combination of internal and external efforts.

Furthermore, effective integration should be conducted for target companies after implementing the M&A strategy. This includes integrating knowledge, technology, personnel, and culture. Communication and collaboration between the R&D departments of both parties should be strengthened, and investment in research and development and in the training of R&D personnel should be increased. The joint research and development mode and the intensity of R&D investment will impact the re-innovative capability pertaining to new knowledge and technology. Companies should comprehensively utilize the R&D capabilities of the acquired party and implement proactive measures to integrate their research institutions and personnel. Internal research and development investment should be prioritized to retain key technical talent; thus, the enhancement of the company's ability in technology transfer and application is enabled. Cultural exchange between companies should be strengthened, and personnel integration should be achieved through training programs and fair reward mechanisms, ensuring that resource integration promotes technological innovation levels in new energy companies.

(2) To stimulate the green technological innovation effect occasioned by the implementation of M&A by new energy companies and to effectively reduce pollutant emissions, measures pertaining to the formulation and implementation of M&A strategies should be implemented, and the following two aspects should be considered:

First, in the process of implementing M&A strategies, new energy companies should focus on their research and development capabilities and technological reserves. This involves cultivating core technical talents, solidifying knowledge foundations, and enhancing knowledge absorption capabilities, which lay a crucial foundation for promoting green technological innovation through resource integration. Based on the M&A context, companies should also enhance their research and development capabilities, particularly in regard to their development and research abilities. Companies should focus on developing the development capabilities of the entire team, which requires prioritizing activities such as concept generation, design, research and development, and testing; thus, the creation of new products and the formation of new processes can be facilitated. A strong research and development capability enables companies to conserve internal resources and mitigate operational risks, and a rich technological reserve enhances the company's knowledge repository, which creates favorable conditions for M&A in the new energy sector. It also reduces the likelihood of post-merger technology integration and transformation failures, which provides a robust foundation for the innovation activities of new energy companies. Therefore, new energy companies should not only actively acquire external resources but also internally prioritize the cultivation of their research and development capabilities and the accumulation of technological resources by increasing research and development investment and expanding the breadth and depth of their knowledge base. Simultaneously, new energy companies should engage in planned and purposeful investigations and experimental activities; thus, they can acquire new knowledge, enhance absorption capabilities, and strengthen their innovation capabilities.

Second, when formulating merger and integration plans, new energy companies should prioritize the implementation of technology-focused M&A strategies. Through technology-driven M&A, new energy companies can quickly acquire assets such as target companies' technological resources and research and development teams, which can supplement, update, and expand the company's knowledge base; shorten the research and development process; overcome technological limitations; and enhance both technological innovation and green technological innovation levels. Therefore, for publicly listed companies in the lithium battery industry, M&A activities should not only focus on vertical integration with upstream suppliers to secure the supply of raw materials but also strongly emphasize M&A related to new materials and new technologies. In the long run, the innovation synergies achieved through technology-driven M&A are key to maintaining a company's market competitiveness. Lithium battery companies should closely monitor the latest technological trends within the industry and pursue collaborative innovation through technology-driven M&A.

(3) After implementing M&A strategies, the elevation of green technological innovation levels in new energy companies is more conducive to reducing pollutant emissions. To stimulate the green technological innovation effects of M&A, government departments should strengthen the protection of intellectual property rights and raise awareness through which new energy companies can defend their rights. The development of a new technology or patent requires substantial investment in research and development, and the process of technology commercialization often exhibits a long time span, which subjects it to various challenges and dilemmas. If new energy companies do not possess a robust mechanism for protecting intellectual property rights to support their research and development achievements, it not only diminishes their enthusiasm for innovation but also increases the risk of core technology leakage and escalates the cost of social rights protection. Frequent disputes over intellectual property rights can disrupt the economic market order. Therefore, on the one hand, the government should implement actions to protect intellectual property rights, intensify efforts to combat infringement and increase penalties for infringers. On the other hand, they should enhance society's overall awareness of intellectual property protection, expand and deepen educational campaigns, and stimulate the enthusiasm of the entire new energy industry for technological innovation and the development of green technologies.

This study exhibits certain limitations. The technological innovation level and green technological innovation level of new energy companies mainly encompass aspects such as novelty, value, and technicality. Merely utilizing the number of technological innovations and green technological innovations effected by companies to represent the innovation level does not facilitate a deeper investigation into the heterogeneous impact of M&A on the technological innovation levels of new energy companies. Additionally, due to the lack of data on M&A scale variables in the lithium battery industry, the "inverted U" relationship between M&A scale and technological innovation was not discussed herein, and further research is required in this research field. Furthermore, the decision-making process for technological innovation in new energy companies is highly complex, and it is influenced by various factors, including the government, the market, and the companies. This study mainly utilized fixed effects, the difference-in-differences models, and control variables to reveal the causal relationship between M&A and technological innovation, which provides credible empirical evidence. However, if natural experiments or instrumental variables, which exhibit critical utility for M&A, could be discovered, an empirical analysis could yield more reliable conclusions.

Author Contributions: Conceptualization, W.C.; methodology, J.Z.; software, J.Z.; validation, J.Z.; formal analysis, J.Z.; investigation, J.Z.; resources, W.D.; data curation, W.D.; writing—original draft preparation, J.Z.; writing—review and editing, W.D.; visualization, H.G.; supervision, W.C.; project administration, W.C.; funding acquisition, H.G. All authors have read and agreed to the published version of the manuscript.

Funding: This study was financed by the National Natural Science Foundation of China, grant number 19BJL040.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: All data generated or analyzed during the current study are presented in this article. Data supporting the findings are available from the corresponding author upon reasonable request.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Chesbrough, H.W. *Open Innovation: The Imperative for Creating and Profiting from Technology;* Harvard Business School Press: Cambridge, MA, USA, 2003.
- PricewaterhouseCoopers: In 2022, M&A transactions in China's new energy industry will hit a record high. *China Business News* (*China*) 9 March 2023. Available online: https://search.ebscohost.com/login.aspx?direct=true&db=edsnbk&AN=19033AF8C6A8 A6A0&lang=zh-cn&site=eds-live (accessed on 28 March 2023).
- Horbach, J.; Rammer, C.; Rennings, K. Determinants of eco-innovations by type of environmental impact—The role of regulatory push/pull, technology push and market pull. *Ecol. Econ.* 2012, 78, 112–122. [CrossRef]
- 4. Berrone, P.; Fosfuri, A.; Gelabert, L.; Gomez-Mejia, L.R. Necessity as the Mother of "Green" Inventions: Institutional Pressures and Environmental Innovations. *Strateg. Manag. J.* **2013**, *34*, 891–909. [CrossRef]
- Lin, H.; Zeng, S.X.; Ma, H.Y.; Qi, G.Y.; Tam, V.W.Y. Can political capital drive corporate green innovation? Lessons from China. J. Clean. Prod. 2014, 64, 63–72. [CrossRef]
- 6. Huang, J.W.; Li, Y.H. Green innovation and performance: The view of organizational capability and social reciprocity. *J. Bus. Ethics* **2017**, *145*, 309–324. [CrossRef]
- 7. Barker, V.L., III; Vincent, L.; Mueller, G.C. CEO Characteristics and Firm R&D Spending. Manag. Sci. 2002, 48, 782–801. [CrossRef]
- 8. Malhotra, S.; Reus, T.H.; Peng, C.Z.; Roelofsen, E.M. The Acquisitive Nature of Extraverted CEOs. *Adm. Sci. Q.* 2018, 63, 370–408. [CrossRef]
- 9. Benischke, M.H.; Martin, G.; Glaser, L. CEO equity risk bearing and strategic risk taking: The moderating effect of CEO personality. *Strateg. Manag. J.* 2019, 40, 153–177. [CrossRef]
- 10. Lee, S.; Oh, W.-Y.; Chang, Y.K. What's inside the Mind of a CEO? The Effects of Discretionary Slack Resources on R&D Investment. *Behav. Sci.* 2023, 13, 247. [CrossRef]
- 11. Granstrand, O.; Sjölander, S. The Acquisition of Technology and Small Firms by Large Firms. *J. Econ. Behav. Organ.* **1990**, 13, 367–386. [CrossRef]

- 12. Ahuma, G.; Katila, R. Technological acquisitions and the innovation performance of acquiring firms: A longitudinal study. *Strateg. Manag. J.* **2001**, *22*, 197–220.
- 13. Cohen, W.M.; Levinthal, D.A. Absorptive Capacity: A New Perspective on Learning and Innovation. *Adm. Sci. Q.* **1990**, 35, 128–152. [CrossRef]
- 14. Cassiman, B.; Colombo, M.G.; Garrone, P.; Veugelers, R. The impact of M&A on the R&D process: An empirical analysis of the role of technological- and market-relatedness. *Res. Policy* 2005, *34*, 195–220. [CrossRef]
- Hitt, M.A.; Hoskisson, R.E.; Johnson, R.A.; Moesel, D.D. The Market for Corporate Control and Firm Innovation. *Acad. Manag. J.* 1996, 39, 1084–1119. [CrossRef]
- Hall, B.H.; Lerner, J. The financing of R&D and innovation. In *Handbook of the Economics of Innovation*; Elsevier: Amsterdam, The Netherlands, 2009; pp. 1–55.
- 17. Hong, X.; Chen, Q. Research on the Impact of Merges and Acquisition Type on the Performance of Listed Agricultural Enterprises—An Analysis of Mediator Effect Based on R&D Input. *Sustainability* **2022**, *14*, 2511. [CrossRef]
- 18. Zhu, H.; Zhu, Q. Mergers and acquisitions by Chinese firms: A review and comparison with other mergers and acquisitions research in the leading journals. *Asia Pac. J. Manag.* **2016**, *33*, 1107. [CrossRef]
- 19. Denicolò, V.; Polo, M. Duplicative research, mergers and innovation. Econ. Lett. 2018, 166, 56–59. [CrossRef]
- Sun, H.; Long, Y.; Yuan, R. Technology M&A and enterprise innovation performance–knowledge-based mediation effect. *Technol. Anal. Strateg. Manag.* 2022, 2022, 1–13. [CrossRef]
- Ma, C.; Liu, Z. Effects of M&As on innovation performance: Empirical evidence from Chinese listed manufacturing enterprises. *Technol. Anal. Strateg. Manag.* 2017, 29, 960–972.
- Liu, X.; Cai, L.; Tan, H. International technology mergers & acquisitions and raising the competitiveness of China equipment manufacturing industry. *Technol. Invest.* 2012, *3*, 17452.
- Huang, L.; Wang, Y.; Shang, L.; Guo, Y.; Porter, A.L. Evaluating the innovation performance of technology mergers and acquisitions in the equipment manufacturing industry. In Proceedings of the Picmet 14 Conference: Portland International Center for Management of Engineering & Technology Infrastructure & Service Integration, Kanazawa, Japan, 27–31 July 2014; IEEE: Piscataway, NJ, USA, 2014.
- 24. Wu, X.; Wang, D.; Yang, T.; Yao, X.; Zhang, H.; Shao, M.; Zhang, W.; Liu, D. Lithium Mining Industry Innovation and Disruptive Technology under the Goal of Carbon Neutrality. *Multipurp. Util. Miner. Resour.* **2022**, *2*, 1–8. [CrossRef]
- 25. Monge, M.; Gil-Alana, L.A. Lithium industry in the behavior of the mergers and acquisitions in the US oil and gas industry. *Energy Sources Part B-Econ. Plan. Policy* **2018**, *13*, 392–403. [CrossRef]
- 26. Monge, M.; Gil-Alaña, L.A.; Cristobal, E. Mergers and acquisitions in the lithium industry: A fractional integration analysis. *Rev. Dev. Financ. Bellville* **2020**, *10*, 31–37.
- Monge, M.; Cristobal, E.; Gil-Alana, L.A. How lithium prices affect mergers and acquisitions in the lithium industry. *Rev. Dev. Financ. J.* 2021, 11, 26–34.
- 28. Sun, Z.; Li, Y.; Wang, M.; Wang, X.; Pan, Y.; Dong, F. How does vertical integration promote innovation corporate social responsibility (ICSR) in the coal industry? A multiple-step multiple mediator model. *PLoS ONE* **2019**, *14*, e0217250. [CrossRef]
- Zhang, Y.; Tong, T.W. How Vertical Integration Affects Firm Innovation: Quasi-Experimental Evidence. Organ. Sci. 2021, 32, 455–479. [CrossRef]
- Lin, D.; Zhao, X.; Hao, T. M&A Size, Technological Absorptive Ability and Post-Acquisition Innovation Performance: Evidence from Chinese High-Tech Firms. In Proceedings of the Asian Academic Accounting Association 18th Annual Conference, Bali, Indonesia, 22–23 November 2018. Available online: https://api.semanticscholar.org/ (accessed on 1 August 2023).
- 31. Hagedoorn, J.; Duysters, G. External Sources of Innovative Capabilities: The Preference for Strategic Alliances or Mergers and Acquisitions. *J. Manag. Stud. Oxf.* 2002, 39, 167–188. [CrossRef]
- 32. Issah, A.-B. Post M&A innovation in family firms. Eur. J. Innov. Manag. 2021, 24, 439–460.
- Jaffe, A.B.; Henderson, T.R. Geographic localization of knowledge spillovers as evidenced by patent citations. Q. J. Econ. 1993, 108, 577–598. [CrossRef]
- Block, J.; Miller, D.; Jaskiewicz, P.; Spiegel, F. Economic and Technological Importance of Innovations in Large Family and Founder Firms: An Analysis of Patent Data. *Fam. Bus. Rev.* 2013, 26, 180–199. [CrossRef]
- 35. Liang, X.; Li, S.; Luo, P.; Li, Z. Green mergers and acquisitions and green innovation: An empirical study on heavily polluting enterprises. *Environ. Sci. Pollut. Res.* 2022, 29, 48937–48952. [CrossRef]
- 36. Li, J.; Wu, X. The S-shaped relationship between R&D investment and green innovation after cross-border merge and acquisition: Evidence from China. *Environ. Sci. Pollut. Res.* **2022**, *29*, 55039–55057. [CrossRef]
- 37. Lv, K.; Pan, M.; Huang, L.; Song, D.; Qian, X. Can intellectual property rights protection reduce air pollution? A quasi-natural experiment from China. *Struct. Change Econ. Dyn.* **2023**, *65*, 210–222. [CrossRef]
- Yin, K.; Liu, L.; Gu, H. Green Paradox or Forced Emission Reduction—The Dual Effects of Environmental Regulation on Carbon Emissions. *Int. J. Environ. Res. Public Health* 2022, 19, 11058. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.